Author's response to reviews

Title: Pingchuan Formula Improves Asthma via Restoration of the Th17/Treg Balance in a Mouse Model

Authors:

- Fei Liu (liufeicindy@163.com)
- Jianer Yu (liufeicindy@126.com)
- Li Bai (1248@szy.sh.cn)
- Zheng Xue (xz695@163.com)
- Xinguang Zhang (1110@szy.sh.cn)

Version: 2
Date: 5 December 2014

Author's response to reviews:

Dear editor#

We have revised our manuscript to include line and page numbers, and included details in our manuscript on the exact constitution of the Pingchuan Formula used in our study.